Author's response to reviews

Title: Ascites due to right atrial myxoma in a haemodialysis patient

Authors:

Anindya Banerjee (anindya.banerjee2@nhs.net)
Andrew Davenport (andrew.davenport@royalfree.nhs.uk)

Version: 4 Date: 2 February 2006

Author's response to reviews:

The additional suggested alterations have now been made. The patient's wife was made aware that this article would be published and consent obtained thereafter.